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Fig. 1 Schematic diagram of porous ceramic membrane preparation under PCS enhancement
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Fig.5 Surface and cross-sectional SEM images of ceramic green membranes prepared using water-ethanol solutions

[ from left to right, water volume fraction: 100%(a), 75%(b), 50%(c), 25%(d), 0% (e)] as coagulation baths
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Table 2 The comparison of O/W emulsions separation performance using ceramic membranes reported in the literatures
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Preparation of polycarbosilane-induced composite ceramic
membranes and oil-water separation
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LI Xing®, LIU Huaizhu®, LIAO Mingjia®*, GONG Genghao'

(1. State Key Laboratory of Advanced Separation Membrane Materials, School of Material
Science and Engineering, Tiangong University, Tianjin 300387, China; 2. College of Chemical
Engineering, Chongqing Chemical Industry Vocational College, Chongqing 401228, China;

3. Tangshan Jiyou Ruifeng Chemical Limited Company, Jidong Oilfield , Tangshan 063500, China)

Abstract; To address the critical challenges of complex fabrication processes, high energy consumption,
and limited separation performance in conventional ceramic membranes, this study proposed a novel
composite ceramic membrane fabrication strategy based on synergistic optimization of polycarbosilane
(PCS)-derived ceramics and phase inversion/sintering processes. Using polysulfone (PSf) as the polymer
matrix, alumina (Al,O;) as the base material, and PCS as the ceramic precursor, a casting solution by
optimizing the N-methyl-pyrrolidone/tetrahydrofuran (NMP/THF) mixed solvent system was developed.
During sintering at 1 450 °C, the silica (Si0);) generated from PCS pyrolysis reacts in situ with Al,O; to
form mullite (3AL; O3 * 2Si0,) crystals, which significantly enhanced the microstructure and mechanical
properties of the membrane material. This study systematically investigated the effects of solvent
composition and coagulation bath conditions on membrane performance. The results demonstrated that the
composite ceramic membranes prepared under various coagulation bath conditions exhibited the following
outstanding advantages: porous structure characteristics with high porosity (52% ~73%) and excellent
compressive strength (18, 5~ 38. 2 MPa), along with tunable pore sizes (90 ~ 320 nm) ; exceptional
separation performance for 1 000 mg/L oil-in-water emulsions [ permeance: 13 240 ~20 540 L/(m?«h+MPa),

(TEE 41 7])
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Molecular dynamics simulation study of water transport
in boron nitride nanotube membranes

WU Jinwen', ZHU Zhihao', KANG Guodong®, CAO Yiming*

(1. School of Chemistry and Chemical Engineering, Anqging Normal University, Anqing 246133, China;
2. Dalian Institute of Chemical Physics, Chinese Academy of Sciences, Dalian 116023, China)
Abstract: Water pollution and scarcity are becoming increasingly severe, driving significant attention to the
development of efficient and sustainable membrane-based water treatment technologies. In recent years,
one-dimensional nanotubes have emerged as a research hotspot due to their exceptional water transport
properties. In this study, molecular dynamics (MD) simulations were employed to investigate the water
transport characteristics in different boron nitride nanotubes (BNNTs) by analyzing the number density,
structure and orientation of water molecules. The simulation results showed that, under the same
conditions, the water flux of the zigzag BNNT (19, 0) with nitrogen atoms as the inlet end was the
smallest. Water molecules will undergo irregular orientation inversion within BNNTs and promote
transport by increasing the bond angle. Furthermore, in nanochannels with different pore sizes, smaller
pore diameters required water molecules to adopt larger average orientation angles and bond angles to
enhance transport. And under certain conditions, long membranes were more conducive to water
transmission. Simulations were conducted over a temperature range of 280 K to 360 K, higher
temperatures reduced the accumulation of water molecules at the entrance, thereby enhancing water

transport, In addition, increasing the pressure also raised the water flux.

Key words: boron nitride; water transport; orientation; molecular dynamics
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due to their ordered pore structure and high hydrophilicity. In this study, a dense seed layer was achieved
by rubbing a pre-synthesized LTA seed sol onto a 200 mm macroporous «—Al, Os; support, followed by the
fabrication of high-performance tubular IL.TA zeolite membranes via a secondary growth method. The
effects of synthesis parameters were systematically investigated to optimize the LTA zeolite membrane
morphology and separation performance, including the presence or absence of a PTFE liner in the
autoclave, water-to-silica ratio in the seed sol, crystallization time, aging time of the synthesis solution,
and rotation speed of the oven. Under optimal conditions, the tubular .TA zeolite membrane exhibited a
high flux of 2. 23 kg/(m’ + h) and an ultrahigh separation factor of 22 050 for dehydrating 90% (mass
fraction) ethanol/water at 348. 15 K. These results highlight the exceptional potential of as-synthesized
LTA membranes for scale-up industrial applications in solvent dehydration processes.

Key words: [LTA zeolite membrane; pervaporation; ethanol dehydration; seed sol-assisted method
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oil rejection; 98. 1% ~99. 4% |; and remarkable antifouling properties, achieving 90% flux recovery
through simple chemical cleaning. Compared to pure Al,O; membrane, the PCS—Al,O; composite ceramic
membrane fabricated under optimized conditions exhibited a 54 % increase in porosity while its compressive
strength did not significantly degrade, maintaining mechanical performance comparable to that of pure
Al,O; membranes, successfully overcoming the technical challenge in conventional ceramic membranes of
balancing high porosity with high mechanical strength.

Key words: ceramic membrane; polycarbosilane; oil-water separation; phase inversion/sintering



